Author's response to reviews

Title: Physical exercise at the workplace prevents deterioration of work ability among healthcare workers: Cluster randomized controlled trial

Authors:

Markus D Jakobsen (markusdue@gmail.com)
Emil Sundstrup (esu@nrcwe.dk)
Mikkel Brandt (mbp@nrcwe.dk)
Kenneth Jay (kan@nrcwe.dk)
Per Aagaard (paagaard@health.sdu.dk)
Lars L Andersn (lla@nrcwe.dk)

Version: 2  Date: 9 June 2015

Author's response to reviews:

Dear Editor,

We would like you to consider the attached manuscript entitled “Physical exercise at the workplace prevents deterioration of work ability among healthcare workers: Cluster randomized controlled trial” for publication in BMC Public Health. The manuscript has not been published elsewhere, and there are no conflicts of interest. All authors have reviewed and approved the manuscript before submission. I (the corresponding author) give full access to all aspects of the research and writing process, and take final responsibility for the paper.

We suggest the following reviewers for their expertise in the field:
Ranta, Radu. Université de Lorraine, CRAN, UMR 7039, 54500 Vandoeuvre-lès-Nancy, France CNRS, CRAN, UMR 7039, France.
radu.ranta@univ-lorraine.fr

Annemarie Lyng Svensson, Department of Occupational and Environmental Medicine, Bispebjerg University Hospital, Copenhagen, Denmark. Tel: +45 35316074; fax: +45 35316070; e-mail: ah46@bbh.regionh.dk

karen.oudehengel@tno.nl

Sincerely,
Markus Due Jakobsen
National Research Centre for the Working Environment,
Lersø Parkalle 105
2100 Copenhagen Ø
Denmark